Author’s response to reviews

Title: Protocol and baseline data for a Prospective open-label explorative randomized single centre comparative study to determine the effects of various intravenous iron preparations on markers of oxidative stress and kidney injury in chronic kidney disease (CKD). (IRON-CKD)

Authors:

Ahmed Zeidan (mda05az@doctors.org.uk)

Sunil Bhandari (Sunil.Bhandari@hey.nhs.uk)

Version: 2 Date: 12 Jan 2019

Author’s response to reviews:

We have now revised Figure 2 according to the reviewer and agree with their comments